Author's response to reviews

Title: High-risk Human Papillomavirus Infection is associated with Premature Rupture of Membranes

Authors:

Geum Joon Cho (md_cho@hanmail.net)
Kyung-Jin Min (mikji78@gmail.com)
Jin-Hwa Hong (jhblue5@naver.com)
Jae-Kwan Lee (jklee38@gmail.com)
Min-Jeong Oh (mjohmd@korea.ac.kr)
Hai Joong Kim (haijkim@korea.ac.kr)

Version: 2 Date: 16 January 2013

Author's response to reviews:

According to your recommendation, we included an ‘Authors’ contributions’ section by suggested format before the Acknowledgements and Reference list. And we comment ethics statement including the name (Korea University Guro Hospital IRB) of the ethics committee that approved our study in last sentence of 1st paragraph, 5 pages.